Author's response to reviews

Title: Study protocol: Cost-effectiveness of two strategies to implement the NVOG guidelines on hypertension in pregnancy: An innovative strategy including a computerised decision support system compared to a common strategy of professional audit and feedback

Authors:

- Susanne H.E. Luitjes (s.luitjes@vumc.nl)
- Maurice C.A.J Wouters (mgaj.wouters@vumc.nl)
- Arie Franx (ariefranx@planet.nl)
- Hubertina C.J. Scheepers (hcj.scheepers@mumc.nl)
- Huub Wollersheim (h.wollersheim@iq.umcn.nl)
- Eric A.P. Steegers (eap.steegers@emc.nl)
- Martijn P Heringa (m.heringa@umcutrecht.nl)
- Veerle M.H. Coupe (v.coupe@vumc.nl)
- Rosella P.M.G. Hermens (r.hermens@iq.umcn.nl)
- Maurits W van Tulder (maurits.van.tulder@falw.vu.nl)

Version: 2 Date: 7 June 2010

Author's response to reviews:

Addition: This paper has not been reviewed previously.